



Adaptive Transcutaneous Power Transfer to Implantable Devices: A State of the Art Review

Kara N. Bocan and Ervin Sejdić *

Department of Electrical and Computer Engineering, Swanson School of Engineering, University of Pittsburgh, Pittsburgh, PA 15213, USA; knb12@pitt.edu

* Correspondence: esejdic@ieee.org; Tel.: +1-412-624-0508

Academic Editor: Vittorio M. N. Passaro Received: 5 January 2016; Accepted: 11 March 2016; Published: 18 March 2016

Abstract: Wireless energy transfer is a broad research area that has recently become applicable to implantable medical devices. Wireless powering of and communication with implanted devices is possible through wireless transcutaneous energy transfer. However, designing wireless transcutaneous systems is complicated due to the variability of the environment. The focus of this review is on strategies to sense and adapt to environmental variations in wireless transcutaneous systems. Adaptive systems provide the ability to maintain performance in the face of both unpredictability (variation from expected parameters) and variability (changes over time). Current strategies in adaptive (or tunable) systems include sensing relevant metrics to evaluate the function of the system in its environment and adjusting control parameters according to sensed values through the use of tunable components. Some challenges of applying adaptive designs to implantable devices are challenges common to all implantable devices, including size and power reduction on the implant, efficiency of power transfer and safety related to energy absorption in tissue. Challenges specifically associated with adaptation include choosing relevant and accessible parameters to sense and adjust, minimizing the tuning time and complexity of control, utilizing feedback from the implanted device and coordinating adaptation at the transmitter and receiver.

Keywords: wireless power transfer; transcutaneous energy transfer; implantable medical devices; adaptive; tuning

1. Introduction

Implantable medical devices have become a huge market, with over 20 million individuals estimated to have an implanted medical device and over \$300 billion in associated costs in the U.S. in 2000. Over one million patients in the U.S. have cardiac pacemakers; 250,000 new pacemakers are implanted each year; 100,000 implantable cardioverter defibrillators (ICDs) are implanted each year; and 120,000 patients in the U.S. have cochlear implants [1].

The term "implantable medical device" has been used to encompass many devices, from pacemakers to orthopedic implants to heart valves. For the purposes of this work, "implantable medical device" will refer to any implanted device that requires electrical power: for example, to stimulate the atrial node in the case of a pacemaker or to power a small microprocessor in the case of an implanted sensing device.

Fully implantable medical devices have already effected vast improvements in patient monitoring and treatment by eliminating percutaneous cables that are prone to infection and limit patient mobility. These include continuous therapy devices, such as the implantable cardioverter/defibrillator (ICD), electronic pacemaker, implantable neurostimulators and fully-implantable drug delivery pumps [2–6].

The benefits of implantable devices also include prostheses, such as the cochlear implant and retinal implant [7–9]. Cortical implant research promises neural control of movement prosthetics and

the valuable inclusion of sensory feedback [1]. Implantable electrodes have been developed for nerve and muscle stimulation [1].

Implantable sensing devices are particularly important in situations where the biological signals to be accessed are inside the body and cannot be reliably sensed non-invasively [10]. This includes wireless implantable biosensors, such as those used with insulin pumps, although commercial sensors are tethered to facilitate removal [11].

Functional challenges associated with fully-implantable devices include powering the implanted device and monitoring or changing settings on the device. Commercial pacemakers and defibrillators rely on non-rechargeable batteries, where the battery is the determining factor in the size of the implant, and periodic surgeries are necessary to replace the device due to the limited battery lifetime (every 5–8 years for pacemakers, every 3–5 years for ICDs) [1].

Depending on the application, implantable devices demand varying levels of power. For example, a pacemaker requires on the order of 10 μ W–1 mW, while a retinal prosthesis requires approximately 45 mW, while a ventricular assist device (VAD) requires 5–25 W, as illustrated in Figure 1 [12–15]. Commercially available VADs still require percutaneous drive lines and an external battery pack due to their high power requirements [1,16]. There is also a considerable range of device sizes, with the VAD being much larger than a retinal prosthesis due to its functional requirements. A third consideration in addition to power level and size is whether a device needs continuous power. Interruption of power will impede the function of any device, but the consequences vary in severity. While interruption of power to a prosthesis will cause a decrease in quality of life and potentially secondary safety issues due to loss of sensory information, the case of a VAD power interruption is immediately life threatening. Meanwhile, an implanted biosensor may only need intermittent power to perform a sensor reading. These device considerations determine which powering methods are feasible for an implanted device.



Figure 1. Range of power requirements of example implantable medical devices.

1.1. Wireless Implantable Devices

Reviews and discussions of implantable device powering methods are provided in [14,17,18]. A focused comparison of inductive and ultrasonic energy transfer is provided in [19], and a review of acoustic energy transfer is given in [20]. Energy harvesting methods utilizing temperature gradients or piezoelectric materials have been developed, but as of yet cannot provide sufficient power for functioning implantable devices [15,17]. While the optimal powering method ultimately depends on the application, electromagnetic (EM) energy transfer has proven to be a promising wireless powering method with a broad range of system designs demonstrated in the literature, capable of delivering varying levels of power depending on size constraints of the device and the required function. Therefore, the focus of this review is on electromagnetic transcutaneous energy transfer.

Transcutaneous EM energy transfer has already enabled improvements to implantable devices by providing a method of powering an implantable device while reducing the dependence on implanted batteries and enabling remote communication with the implant. The cochlear implant was the first commercial wirelessly powered implantable device [7,8,21,22]. Wireless power is provided to the cochlear implant through several millimeters of tissue, along with communication of processed auditory information from an external microphone. Retinal prostheses are a recent development and have been successfully implanted in humans to restore sight [9]. There have also been research

efforts towards a fully implantable wirelessly powered VAD, even with the challenging power requirements [16].

Wireless transcutaneous systems generally include an external antenna and an implanted antenna, as illustrated in Figure 2. EM energy is transmitted through the tissue from the external antenna, captured by the implanted antenna, and rectified to power implanted circuitry. The energy captured at the implant can be utilized to recharge a battery or to directly power a battery-less implanted device. Eliminating the implanted battery typically enables miniaturization of the implant, improving patient safety and comfort by lowering infection risk and lessening the obtrusiveness of the implant [1]. Depending on the power requirements and the need for continuous power, the power source of an implantable device can be primarily wireless, with a battery to provide backup power for critical functions, thereby prolonging the lifetime of the implant while providing a fail-safe for power interruptions [23].

Modulation of transcutaneous energy can be performed to achieve communication between the external and implanted sides. Back communication (implant to external) can be accomplished using a transmitter on the implant (active), or by modulating the energy from an external transmitter (passive), with active communication generally requiring more power at the implant. While the focus of this review is on electromagnetic energy transfer, similar communication has been achieved with other wireless powering methods such as ultrasound [24–26]. Wireless transcutaneous communication allows control of the implanted device behavior as well as access to information from the implant, such as readings from implanted sensors. This enables programmability of implanted devices and opportunities for post-surgical or remote long-term monitoring by reading information from the implant.



Figure 2. Simplified general wireless transcutaneous system architecture.

The main challenge associated with wireless transcutaneous energy transfer is achieving sufficient and reliable energy transfer to power the implant within safety limits on absorption in tissue. Electromagnetically, tissue is a lossy dielectric material with properties described by permittivity and conductivity. EM energy absorbed in tissue causes heating, hence there are fundamental safety concerns involved in transcutaneous energy transfer.

The goal of maximizing energy transfer to the implant while minimizing energy absorption in tissue has logically led to definitions of efficiency in terms of the power delivered to the implanted load, absorbed power, and input power [27–29]. The goal in the design of a system then becomes to maximize this efficiency. Conceptually, efficiency represents a distribution of input power in the source, antennas, system components, and load [30]. A more specific definition of efficiency for a particular system depends on multiple parameters, including: Antenna topologies, antenna dimensions and separation, the properties of the media surrounding the antennas, and operating frequency. These parameters determine the characteristics of the electromagnetic fields in the system. Once the system has been

characterized, there are numerous strategies for optimizing the efficiency, whether through frequency tuning, impedance matching, or load tuning.

1.2. Environmental Variations

While a transcutaneous system can be optimized for a particular configuration, wireless transcutaneous energy transfer in practice is complicated by unpredictability and variability of the physiological environment. Variations occur due to antenna misalignment, movement of the antennas or the patient, implant migration, and changes in tissue structure and properties. Changes in the environment directly affect the performance and safety of a transcutaneous system, through effects on impedance and field characteristics. Impedance changes can reduce power transfer at the transmitter and receiver, as well as degrading efficiency if the impedance deviates from an optimal load or if there is increased absorption in tissue [31–33]. The degradation in power transfer and efficiency can lead to other problems in the system: Power reflections at the transmitter can damage the system, while reduced power delivery to the receiver can cause interruptions in function, and excess power delivery to the receiver circuitry or lead to tissue heating and safety concerns [34–38].

To avoid or lessen the degradation in system performance, sources of environmental variations must be examined and accommodated. One source of variation is changes in positioning or alignment of the antennas, whether due to movement of the patient or implant migration. Inductively coupled systems are particularly sensitive to variations in distance or alignment [39]. Some systems employ flexible antennas on the tissue surface, where movement can change the antenna geometry as well as the alignment [40]. In retinal prostheses, the implant antenna is in near constant movement due to its location in the eye [41]. Migration of an implant can occur after implantation, resulting in uncertainty of the implant location and potential misalignment of the antennas [42]. Distance or alignment variations can cause reduced power delivery and interruption of the implant function [43].

Although less attention is devoted to potential changes due to differences in tissue characteristics, there is evidence that tissue varies among patients and over time [44–51]. These variations are in addition to the known frequency dependence of tissue dielectric properties. Changes in body chemistry (such as hydration or fat content), inflammation, fibrous encapsulation, and changes in cellular structure are well-documented physiological processes that affect tissue properties. Changes in tissue thickness will result in variable separation between an external and implanted antenna, causing either increased coupling due to thinner tissue or greater attenuation of the fields through thicker tissue. Changes in tissue structure or tissue chemistry, and associated changes in tissue properties, will affect the antenna electrical size and impedance as well as the optimal frequency of operation.

Attempts to minimize the effects of variations have been realized through design of insensitive systems and adaptive systems. Insensitive systems are designed for consistent operation over a range of environmental parameters, without the use of tuning components [35,52]. Adaptive systems include tunable components to adjust the characteristics of the system in response to changes in the environment. The focus of this review is adaptive systems specifically designed for transcutaneous operation in the electromagnetic near or mid-field.

1.3. Review Organization

There is a wealth of literature on wireless power transfer, energy harvesting, antenna design, and tissue electromagnetics. Design concepts and challenges from each of these fields are combined in the design of wireless implantable medical devices. As such, there is much that cannot practically be covered in the background for this review. Although there are significant biomaterials and biocompatibility challenges associated with implantable medical devices, the focus of this work is on electromagnetic wireless transcutaneous powering of implantable devices. Additionally, the literature on electromagnetic wireless power transfer covers many components of system design, including power amplifier design, rectifier design, antenna design and voltage regulation. These components are essential and have been studied in terms of improving efficiency, and background information

is provided in this work where relevant. Similarly, this work will address but not focus on aspects of wireless communication in transcutaneous operation, instead maintaining the focus on adaptive methods for transcutaneous wireless power transfer. The background of this review is meant to provide a general overview of concepts relevant to transcutaneous power transfer and the adaptive methods covered later in the paper, consolidating information from the many fields of research that contribute to transcutaneous powering of implantable devices.

The paper is organized as follows: Section 2 covers background topics relevant to the review of the literature on adaptive transcutaneous systems; Section 3 provides an overview of the design process for an adaptive system and the strategies that have been implemented or proposed for each stage of the process; Section 4 details implementations in the literature specifically for implantable devices and transcutaneous power transfer; finally, Section 5 provides a summary of the literature and a discussion of remaining challenges in the field.

2. Background

2.1. Wireless System Architecture

A wireless system typically consists of the components shown in Figure 2. At the primary or transmitter (external), the ac transmit signal is generated and fed to the transmit antenna. The EM field generated by the transmit antenna is captured by the receive antenna and converted to a dc voltage to power the receiver circuitry.

Oscillators and power amplifiers are common at the transmitter to generate the ac transmit signal. The power amplifier class and amplifier efficiency are design concerns at the transmitter. A matching network is often included to match the transmit antenna to the optimal load of the power amplifier to improve efficiency. At the receiver, rectifier efficiency is a design concern, as well as voltage regulation to supply a stable voltage to the receiver circuitry. The choice of transmit and antenna topology and dimensions depends on the application.

2.2. Field Regions

Wireless systems can operate in various field regions, depending on the operating frequency, antenna size, antenna separation and transmission medium. Field regions are characterized by the types of fields present. The reactive near field is closest to the radiating antenna and contains primarily "stored" energy; the radiating near field contains both radiating and reactive fields with radiating fields dominating; the far field contains radiating electric and magnetic fields in planes transverse to the direction of propagation and the angular field distribution can be considered independent of distance from the radiating antenna [53,54].

Definitions of the boundaries between the fields vary and the transition between regions is gradual, but the region boundaries can be approximated in terms of the electrical size of the antennas and the electrical distance between them (relative to the wavelength at the operating frequency) [53,54]. The electromagnetic field region determines the interaction (if any) of the transmitting and receiving antennas, which guides the associated design and analysis of the system. Theoretically, any antenna system can be made to operate in any field region by varying the size or separation of the antennas. However, the efficiency of a system is expected to degrade when operating outside of the desired field region to the point where the system may cease to function.

Systems designed to operate in the reactive near field include inductively coupled or magnetically coupled systems, where the magnetic field of a transmitter coil induces a current in a receiver coil [40]. The term "antenna" is not typically used to refer to the coils in inductively coupled systems. Rather, the term "coil" or "resonator" is used to designate that the coils are designed as coupled (reactive) and not as radiating antennas. Capacitive coupling via the electric field is also possible in the reactive near field, and has enabled power transfer to implantable devices in proximity to metallic implants [55]. Efficiency in an inductively coupled system decreases significantly with greater coil separation or

Weakly coupled systems can be designed to have higher efficiency in the radiating near field region by deliberately utilizing differently sized antennas [57]. Such a system can be designed with the receive antenna dramatically miniaturized relative to the transmit antenna. This "midfield" region of operation has been recently investigated for transcutaneous powering of miniature implantable devices.

Far field systems operate at a distance and frequency such that there is no detectable coupling between the transmit and receive antennas [54]. This region is less used for implantable devices to avoid problems with attenuation of propagating fields in tissue. The focus of this review is on transcutaneous systems with an external antenna on or near the tissue surface and an implanted antenna, limiting the discussion to reactive near field and midfield systems.

For transcutaneous applications, inductively or magnetically coupled systems are well suited when the coil separation can be on the order of the coil diameter to maintain high efficiency, and where the external and implanted coils can be of similar size. Such systems have been shown as capable of delivering tens of watts of power to an implanted device at small coil separations [12,43]. In cases where miniaturization of the implanted device is a priority, the separation between the implanted and external antennas is likely to be larger than the antenna dimensions. Therefore, the efficiency will be maximized by designing for midfield operation with asymmetrically sized antennas. Such systems have been shown to achieve power delivery of up to several milliwatts to a millimeter-sized implant [15].

2.3. Tissue Properties

Two main current paths can be identified in tissue, corresponding to permittivity and conductivity and the current paths described in Ampere's Law. One path is conduction current, corresponding to the movement of ions in tissue and therefore the conductivity (σ). The other path is displacement current, a function of the electric flux and permittivity (ϵ).

Conductivity is a function of ionic mobility in the tissue medium, where ions such as sodium and potassium act as charge carriers. Permittivity is a function of charge buildup at cell membranes (cell membrane capacitance) and alignment of molecular dipoles with an applied field [58]. Dielectric relaxation is also included in the complex permittivity, describing the effect of molecular dipole rotation delay in response to an applied field. Tissue conductivity and permittivity are frequency dependent, with conductivity increasing with frequency, and permittivity decreasing with frequency.

In general, conduction current can be utilized at low frequencies and small antenna (electrode) separations, where the movement of ions creates capacitive charge transfer. At higher frequencies or greater antenna separations, conductivity contributes to loss due to field vector directions and the inability of the ions to build up along interfaces. At higher frequencies, displacement current becomes significant, and the induced current due to greater flux competes against conductive losses. At GHz frequencies, dielectric relaxation losses also become significant. The specifics of energy transfer differ based on the tissue structure, thickness, and chemistry. Tissue properties have been measured for multiple tissue types across frequency, and parametric models have been developed to represent tissue properties over a range of frequencies [47,58].

Tissue properties are particularly relevant to wireless power transfer due to their relationship to power dissipation in tissue, which can lead to tissue heating and therefore presents safety issues. Energy absorption in tissue is quantified as specific absorption rate (SAR), measured in watts per kilogram. SAR limits defined by several organizations are used as safety guidelines for transcutaneous energy transfer. These limits have been defined based on studies of physical and behavioral effects of electromagnetic field exposure in animals and humans, but are not specific to medical use. The IEEE Standard for Safety Levels with Respect to Human Exposure to Radio Frequency Electromagnetic Fields 3 kHz–300 GHz (IEEE Std C95.1-2005) specifies limits of 0.4 W/kg and 10 W/kg for whole body and local SAR, respectively [59]. The International Commission on Non-ionizing Radiation Protection (ICNIRP) guidelines specify that general public exposure is limited to 0.08 W/kg whole body, 2 W/kg local for head and trunk, and 4 W/kg local for limbs, with local SAR averaged over 10 g of tissue for any 6 min period [60]. For uncontrolled exposures of the general population, the U.S. Federal Communications Commission (FCC) limits SAR to 1.6 W/kg averaged over any 1 g cube of tissue [61]. In addition to dielectric losses, heating can also occur due to heat dissipation of the receiver circuitry, sometimes exceeding the one-degree Celsius temperature increase that is a basis for SAR limits [41,62,63].

2.4. Operating Frequency

Due to the complex frequency dependent dielectric properties of tissue, the existence of an optimal frequency has been proposed that balances power transfer and losses in tissue [27,28]. The optimal frequency was determined by defining a measure of efficiency in terms of power delivered to the load and power absorbed in tissue [27]. Ho *et al.* [57] showed that the highest efficiency is achievable in the midfield for weakly-coupled disparate-sized antennas, resulting in an optimal frequency in the sub-GHz to GHz range for a cm-size transmitter and mm-size receiver at cm-separation. Higher frequency has also been recently investigated as it enables miniaturization of antennas, greater antenna impedance, and greater open circuit voltage at the implant [28,32,39,57,64].

2.5. Transcutaneous Antennas

Particular antenna topologies tend to be favored for transcutaneous systems due to size constraints and the effects of tissue properties. Miniature antennas are desirable for implants to reduce patient discomfort and infection risk, but this results in electrically small antennas where impedance mismatch can lower efficiency [65]. Miniature antennas have been shown to be most efficient in weakly coupled systems operating in the midfield [27].

Tissue's high permittivity (due to its composition of mostly water) reduces the net electric field, an effect of molecular alignment with an applied field. Loop or coil antenna topologies are widely used in the implantable device literature to take advantage of tissue's non-magnetic permeability and utilize the magnetic near field. Fewer turns have been shown to be better in proximity to conductive high permittivity tissue, with quality factor decreasing with increasing turns [28]. The induced current at a loop receiver depends on the flux through the loop, hence the advantage of using higher frequencies. At frequencies and antenna dimensions such that there is midfield operation, it has been shown that both the electric and magnetic fields contribute to power transfer [66].

Inductive antennas (loops) act as better receivers due to their low impedance, while capacitive antennas act as better transmitters due to their larger impedance which limits current flow at the transmitter [14]. Low impedance can lead to power loss and heat generation due to high currents when used as a transmitter [39].

Printed antennas are favored due to the potential for miniaturization and manufacturability [28]. More complex antenna designs have been investigated to focus fields at the site of the implanted receiver, and to minimize excess power dissipated in tissue [15,42,67].

2.6. Power Gain and Efficiency

Because the function of any receiving device depends on its power supply, it is desirable to maximize power transfer to an implant. However, transcutaneous operation necessitates consideration of safety limitations relating to absorption of electromagnetic energy in tissue. The need to maximize power to the load while minimizing absorption in tissue then evokes definitions of efficiency.

Definitions of efficiency vary in terms of what the load power represents and what it is expressed relative to: power available to the load relative to the power absorbed in tissue [15,27,42]; power to the

load relative to power available from the source, representing the transducer gain [68]; power to the load relative to power delivered by the source, with maximum efficiency achieved with an optimal load [32]; power to the load relative to power input to the network, maximized with a conjugately matched load in a weakly coupled system [57,66]. Optimal load impedances have been derived to maximize efficiency according to the various definitions [29,32,57]. The definitions of efficiency overlap with definitions of power gain, but for passive systems the term efficiency is used to indicate that the power gain from source to load is less than one.

Defining efficiency relative to the power from the source, there exists an upper bound of 50% power efficiency with simultaneous conjugate matching [32]. While the load resistance is equal to the source resistance for maximum power transfer, the load resistance is larger than the source for higher efficiency (relative to the source power) [29]. Designing for maximum efficiency in this case may reduce the range of operation due to reduced power transfer to the load [29]. However, maximizing voltage at the implant is sometimes preferable even at the expense of power transfer, to ensure adequate turn-on voltage while minimizing losses in the system. Additionally, in the case of delivering power from a power amplifier, efficiency (defined as output power from the power amplifier relative to input power) is prioritized to minimize heat dissipation in the circuit. In this case, the power amplifier is designed to operate at maximum efficiency, rather than using conjugate matching, and an optimum load can be defined [32,69].

Defining efficiency as load power relative to power absorbed in tissue, maximum efficiency is achieved through conjugate matching [27,57]. For delivering power to a load through tissue, this efficiency is valuable when power dissipation in tissue is of more significant interest than power dissipation elsewhere in the system.

The system efficiency or gain can be decomposed into component efficiencies, including coupling efficiency that represents the efficiency between the transmit and receive antennas (coils) [70], power amplifier efficiency [32], and rectifier efficiency [71–73]. In midfield cases where losses in tissue are significant and lumped element models are no longer appropriate, network analysis allows representation of a linear system in terms of scattering parameters (S-parameters) to calculate efficiency or power gain [27,69,71,74].

2.7. Impedance Matching

Impedance matching networks can be added to transform the real and reactive impedances looking into portions of a system, to minimize voltage reflections at interfaces, to conjugately match for maximum power transfer, or to achieve optimum loading.

Common network topologies include L, T, and pi networks consisting of inductors and capacitors in configurations capable of increasing or decreasing input impedance. The choice of topology depends on design parameters including the desired bandwidth, impedance transformation range, complexity, and available area [75].

Pi-match networks can be used to both increase and decrease impedance, while L-match networks can only be designed to transform the impedance in one direction (either increase or decrease) [69]. Pi networks can provide wider band matching, but L-match is appropriate when efficiency is of primary concern [32].

2.8. Impedance and Material Properties

Impedance is closely related to material properties of conductivity, permittivity, and permeability, and properties of inductance, capacitance, and resistance. Impedance is typically represented as a function of resistance (R) and reactance (X) as given in Equation (1). Resistance is then a function of geometry and conductivity, while reactance is a function of geometry and permittivity or permeability. Reactance can also be expressed in terms of inductance (L) or capacitance (C), also indicated in Equation (1).

$$Z = R + jX = R - \frac{j}{\omega C} = R + j\omega L \tag{1}$$

2.9. Resonance

Resonance in the context of antenna or circuit design typically refers to a system with completely real input impedance. A natural resonance associated with the system geometry has also been defined as where the determinant of the scattering matrix approaches zero [65]. For the purposes of this review, the focus will be on the first definition of resonance due to its relevance to impedance matching. Resonance tuning is equivalent to impedance matching to achieve an all-real input impedance, implemented to maximize power delivered to or from a real impedance. Designing antennas with inherent resonance can avoid losses associated with an added matching network. Operating transmit and receive antennas at the same resonant frequency increases the voltage gain between the antennas [13,41].

Near field systems tend to utilize resonant tuning to design transmitting and receiving antennas (coils) to function at the desired operating frequency. Resonance tuning can include added capacitors or inductors to compensate for an antenna's reactive impedance, or antennas designed such that the inherent capacitance and inductance are tuned to resonance [65]. Magnetic resonance systems have been designed with three or four coils, where the additional coils perform impedance transformation at the transmitter and/or receiver [56,76]. This is assuming that the coils are being designed with a specific frequency in mind. In some cases, the frequency is instead tuned to achieve resonance with the given impedance characteristics of the system. It has been noted that the optimum load differs from resonant tuning if the media between the transmit and receive coils is conductive, as in the case of transcutaneous operation [32].

2.10. Quality Factor

Quality factor (*Q*) is a measure of the energy stored in a system relative to the energy dissipated or lost per time. For a passive parallel RLC network at resonance, *Q* is defined as in Equation (2) [69]. The stored energy (E_{stored}) is the peak energy stored in the capacitance (*C*) or inductance (*L*), as the oscillatory energy is transferred between the two. The power dissipated (P_{avg}) is the power through the resistance (*R*).

$$Q = \omega_0 \frac{E_{stored}}{P_{avg}} = \frac{R}{\sqrt{LC}}$$
(2)

Q represents the ratio of the current flowing in inductors/capacitors to the net current through the network. Therefore, operating at high Q increases the voltage swing across the LC part of the network. Higher Q also corresponds to narrower fractional bandwidth, as shown in Equation (3) [69].

$$\frac{BW}{\omega_0} = \frac{1}{Q} \tag{3}$$

High Q for an antenna equates to low radiation efficiency, and there is a fundamental limit on the minimum Q of electrically small antennas [54]. One strategy for achieving better radiation properties is space-filling or meandering, where an antenna occupies an equivalent area but is made to have an increased electrical length [54]. In the case of near field operation, however, designing electrically small antennas to operate at high Q is desirable to increase the "stored" near field energy. The basis for magnetic resonance systems is designing coil antennas to operate at high Q [12]. Because high Q equates to narrower bandwidth, coupled systems can be more efficiently designed using separate coils (and separate frequencies) for power and data [73].

Q of a material is inversely related to the loss tangent (tan δ) of the material, defined in Equation (4) in terms of conductivity and permittivity.

$$\tan \delta = \frac{1}{Q} = \frac{\sigma}{\omega \epsilon} \tag{4}$$

3. Components of Adaptive Transcutaneous Systems

The goal of adaptation in any system is to maintain desired performance despite environmental variations. How "performance" is defined and the strategies to maintain performance depend on the application and the operating field region of the antennas.

The general steps in the design of an adaptive system are illustrated in Figure 3: (a) Defining a parameter to optimize (a performance metric) based on the functional goal of adaptation; (b) developing a tuning method based on the available controls affecting the performance metric and the expected range of parameter variations; and (c) implementing sensing to determine or evaluate the tuning state. This section will provide a conceptual overview of the strategies employed in each stage of the design, and the following section will cover specific implementations in the literature.

It should be noted that the focus of this review is on adaptive methods for electromagnetic wireless transcutaneous systems. There is a substantial amount of literature on adaptive electromagnetic wireless systems that is outside the scope of the current review, but methods from these areas are included where they have been applied to transcutaneous power transfer to implantable devices.



Figure 3. General process steps in designing an adaptive system.

3.1. Performance Metric

The parameter used as an indication of system performance depends on the desired function of the system and the system characteristics. Strategies include minimizing reflections at interfaces, maximizing power transfer, maximizing efficiency, and maintaining constant load voltage.

Maximum power transfer can be maintained in variable environments through tunable impedance matching. Minimizing reflection achieves maximum power transfer when the goal is matching to an all-real impedance. When matching to complex impedances, maximum power transfer is achieved through complex conjugate matching.

There are various definitions of efficiency, but power efficiency is typically defined as load power relative to input power or power absorbed in tissue [27,32]. Design strategies to maintain efficiency in the presence of environmental changes are many and varied, but most can be generally classified into frequency tuning and impedance tuning.

3.2. Tunable Components

Tunable components are essential to adapting a system with the goal of optimizing a chosen parameter. In frequency tuning applications, a voltage controlled oscillator can be used to adjust the transmitter frequency [40,77,78]. The switching frequency of the power amplifier at the transmitter can also be controlled [73].

In impedance tuning applications, variable inductances and capacitances can be used in matching networks to transform impedances. Transductors provide current-controlled inductance, and can withstand large voltage and current, but they tend to be too bulky for wearable or implantable devices [40,79]. Microelectromechanical systems (MEMS) inductors have been shown to provide high efficiency and tuning range, but variable inductors remain difficult to implement on-chip for miniature implants [80,81]. Variable capacitances tend to be favored over inductances due to the widespread use of inductive coil antennas and the higher quality factor of capacitors on-chip [32,75,78]. Voltage-controlled capacitors (varactors) offer compact area, but they can exhibit non-linearities at radio frequencies and require an analog control voltage [40,64,68,75,82]. Switched capacitor banks have also been used to vary parallel matching capacitance, but their disadvantages include the greater space occupied and the necessary discretization of the capacitance values [76,83–85].

Another approach to impedance tuning is duty cycling. Ahn *et al.* [76] implemented buck-boost converter load connection duty cycling to modulate effective load in an inductively coupled system; Si *et al.* [83] duty-cycled switched capacitors to achieve variable effective capacitance. Variable output dc-dc converters and control of power amplifier supply voltage have been used to regulate supply voltage and input power at the transmitter [37,40]. In some cases the antenna itself can become a tunable component, either through adjusting multiple feed points to achieve beamforming, or through changes in the antenna geometry [15,86,87].

3.3. Sensing and Feedback

Feedback can be implemented to evaluate the state of the performance metric and to determine the appropriate control of tuning components. The various approaches in the literature fall into three categories, illustrated in Figure 4: (1) Sensing and processing at the external transmitter to tune components at the transmitter; (2) sensing and processing at the implanted receiver to tune components at the receiver; and (3) sensing at the implanted receiver communicated to the external transmitter for processing and tuning components at the transmitter.



Figure 4. Existing approaches to adaptive transcutaneous system design.

Parameters used for feedback include the rectified voltage indicating received power [40,88], reflected voltage indicating impedance mismatch [77], and phase differences to detect reactive impedance [40]. Information on parameters sensed at the receiver can be communicated to the transmitter passively by modulation of the power carrier, or actively through the use of an implanted transmitter.

The tuned components, sensed parameter, and the control method are closely related. The control method is designed based on the parameters available to sense, the controllable parameters that affect the optimized parameter and the subsequent choice of tunable components, and any design constraints on tuning time, size, and power consumption.

The tuning time must be controlled such that it does not interfere with primary functions of the system such as communication. An iterative algorithm such as gradient search is relatively simple to implement, but has a disadvantage in tuning time compared to single iteration methods based on direct calculation [80]. However, the power and memory requirements of complex vector calculations can limit the application of such methods on miniature implantable devices. A balance must be achieved among tuning time, complexity, and accuracy.

Adaptation control can be implemented at either the transmitter or receiver, or both the transmitter and receiver. Strategies at the transmitter include adjusting transmitter power and frequency, or tunable matching at the transmit antenna [12,34,36,37,40,43,82,83,89,90]. Adaptation strategies at the receiver mainly include tuning matching networks to achieve optimal load [72,90–92].

Adaptation of both sides of the network is necessary for maximum power transfer efficiency, and may require communication of the adaptation state from one side of the network to the other [64,70]. For any adaptation strategy to extend to an implantable device, it must be miniature and low power to avoid adding significantly to the existing power and size requirements of the implant. There are fewer restrictions on the transmitter in terms of size and power consumption, but safety is still a primary concern when transmitting power through tissue [14].

4. Implementations of Adaptive Transcutaneous Systems

The combination of strategies discussed above depends on which environmental variation(s) are addressed and where in the system the adaptation is performed. A review of implementations in the literature is given in Table 1, indicating the goal of adaptation (including the performance metric), tuning performed and feedback at the external transmitter (Tx) and/or the implanted receiver (Rx), and the environmental variation for which the adaptation is intended to compensate. Designs of interest are categorized and discussed in the following subsections.

| | Carl | Tuning | | Feedback | | Variation |
|---------------|---|--|--------------------------------------|--|--|---|
| | Goai | Tx | Rx | Tx | Rx | variation |
| [77] | Min reflections | Frequency (VCO) | | Reflected voltage | | Resonant frequency |
| [78] | Max efficiency | Frequency (VCO) | | Antenna voltage | | Resonant frequency |
| [40] | Max received voltage Max driver efficiency | Frequency (VCO), Power (supply voltage) Amplifier Z_L (transductor) | | Phase between LC and coil driver voltage | Rectified voltage | Resonant frequency, coupling Impedance due to frequency change |
| [82] | Resonance | Impedance (varactor) | | Reflected voltage phase change | | Distance and impedance |
| [83] | Resonance | Impedance (duty-cycled C) | | Frequency | | Impedance |
| [39] | Stable received power | Power (supply voltage) | | | Rectified voltage | |
| [13] | Stable received power | Impedance (duty-cycled C) | | | Rectified voltage | |
| [36] | Stable received power | Power (supply voltage) | | | Storage capacitor voltage | Movement and load impedance |
| [73] | Max efficiency | Frequency | | Resonator voltage | | Coupling, impedance |
| [80] | PA optimal load | Matching impedance (simulated) | | Antenna impedance | | Impedance |
| [31,89,93] | Match to PA or LNA | Matching impedance (varactor, switched C bank) | Matching impedance (varactor) | Antenna impedance | Antenna impedance | Impedance Impedance |
| [14,41] | Max power transfer efficiency | Power (supply voltage) Impedance | | | Rectified voltage Rectified voltage | Position, fibrous tissue growth |
| [64,71,81,90] | Max power transfer efficiency | Matching impedance (switched C bank) | Matching impedance (switched C bank) | | Rectified voltage Rectified voltage | Position, impedance |
| [37] | Stable received power | Power (supply voltage) | | | Rectified voltage | Distance, alignment |
| [79] | Max received voltage Stable received power | Impedance (switched C bank) Power (supply voltage) | | | Rectified voltage Rectified voltage | Position Position |
| [72] | Max efficiency (optimum load) | | Impedance (switched C bank) | | Rectified voltage gradient | Load impedance |
| [34] | Stable received power | Frequency (ZVS) | | Switch transistor drain voltage | | Coupling, load impedance |
| [15] | Max efficiency (relative to absorption) | Field pattern (antenna feeds) | | | Rectified voltage | Position |
| [92] | Max power transfer | | Impedance (switched CLC pi bank) | | Antenna port voltage | Impedance |
| [43] | Max power transfer | Frequency (ZVS) | | Resonant tank voltage | | Coupling |
| [84] | Max efficiency | Impedance (variable C) | | S-parameters | | Coupling |
| [23] | Max power transfer Max efficiency | Matching impedance (variable C) Power | | S-parameters Reflected voltage | | Coupling Coupling |
| [12] | Stable power, max efficiency | Power, resonant frequency | | | Received power | Coupling |
| [94] | Resonance | Impedance (switched C) | | Output voltage, tuning capacitor voltage | | Resonant frequency |

Table 1. Summary of literature on adaptive transcutaneous systems.

4.1. Input Power Adjustment

Changes in the separation distance and alignment of antennas can cause variations in coupling and power transfer [95]. One strategy to compensate involves regulating input power at the transmitter. The motivation for this tuning is to avoid or reduce the power regulation burden at the implant, which contributes to heat dissipation [39].

Van Schuylenbergh and Puers [40] adjusted the transmit power using a voltage-controlled boost regulator, based on feedback of the dc input voltage at the internal regulator.

Wang *et al.* [36] compensated for coil movement or load changes, regulating transmitted power by adjusting the supply voltage to a power amplifier. The supply voltage was adjusted according to the detected voltage on a storage capacitor at the receiver, communicated to the transmitter over a second frequency band. Power level adjustment and back telemetry were staggered to avoid interference.

Si *et al.* [39] demonstrated power regulation at the transmitter, designed to limit the size and heat dissipation associated with power regulation at the receiver. Power was regulated by adjusting the supply voltage at the primary resonant converter based on feedback on the dc voltage at the receiver, communicated wirelessly to the primary.

Ng *et al.* [14,41] proposed a system to compensate for eye movements and fibrous growth in retinal implants, regulating the supply voltage at the transmitter based on feedback on the voltage at the secondary, communicated through back telemetry. Adjusting the transmitter resonant frequency for maximum efficiency by tuning the transmitter capacitance was also proposed, based on voltage at the receiver. A primarily theoretical discussion was presented for tuning and feedback.

Kiani and Ghovanloo [37] designed a system to compensate for distance and angular alignment variations in an inductively coupled system, regulating transmit power based on an indication of voltage across the receive coil. Back telemetry was performed if the received voltage was greater than a reference value, so the transmitter increased transmit power unless bits were received from the implant. The design focused on using primarily off-the-shelf components.

Waters *et al.* [23] designed a system to adjust transmit power based on reflected voltage at the transmitter to compensate for coupling variations in a magnetic resonance system. They later proposed an auto-tuning algorithm to adjust transmit power based on detected load power, and measured the temperature of the receive coil as a measure of efficiency [12]. Back telemetry of load power was mentioned as future work.

4.2. Adaptive Antennas

In an effort to compensate for changes in alignment of the antennas, which can lead to changes in coupling and power transfer, transmitters have been designed with location and focusing capabilities. The concept is related to power regulation, but the regulation can involve multiple antennas or control of multiple feeds to a single antenna.

McMenamin *et al.* [96] presented a system that adjusted powering of antennas in animal cages for bio-telemetry experiments based on the animal's detected position. The goal was to focus wireless electromagnetic energy to power a mobile telemetry unit worn by the animal. The floor of the animal cage consisted of an array of overlapping planar spiral coils, and the magnetic field within the cage was focused by only providing power to the coils closest to the detected position of the mobile telemetry unit. A small magnetic tracer was embedded in the mobile telemetry unit and an array of magnetic field sensors in the cage floor was used to detect the unit's location.

Ho *et al.* [15] manipulated the field pattern itself to maximize efficiency by implementing a transmitter with dynamic focusing capability. The transmitter was designed to power a miniature (2 mm \times 3.5 mm) implant in the midfield (5 cm separation) for pacemaker or cortical implant applications. Through control of the phases of the antenna feeds, the field was focused at an implant in various locations. The focusing was adjusted using an optical indication of received power as feedback for the purpose of the experiment. A method of back telemetry to practically adjust the focusing was not discussed.

4.3. Frequency Tuning

Frequency tuning has been utilized to correct for changes in antenna impedance due to the surrounding environment, including maintaining operation at a resonance frequency. Frequency tuning can be difficult due to FCC regulations on frequency bands [84,91]. It is mainly used in inductively coupled systems, where the coils can be designed such that the coupled system does not radiate significantly, but the operation is highly affected by coil separation distance [68]. A review of automatic frequency control techniques is provided in [97], and several examples of frequency tuning in transcutaneous systems are presented here.

Ko *et al.* [77] used a voltage-controlled oscillator to adjust the transmit frequency in response to detuning of an animal's cage for bio-telemetry experiments. The goal was to maintain operation at the resonant frequency of the transmitter circuit including the animal's cage, in order to power a battery-less implant for chronic animal telemetry. Detuning due to the animal's movement was detected by measuring reflected voltage at the transmitter, and the transmitter frequency was adjusted to minimize reflections. Phase-lock loop techniques were proposed as a strategy for tuning the receiving unit in response to changes in the transmitter frequency.

Fernald *et al.* [78] also used a voltage-controlled oscillator, and adjusted the transmit frequency in response to changes in the antenna resonant frequency. The application was a general-purpose implant for animal telemetry experiments. The system performed a resonant search, sweeping the frequency and monitoring the voltage amplitude across the transmit antenna. When the voltage swing reached a threshold value, the frequency was fixed and the system began transmitting at that frequency.

Baker and Sarpeshkar [73] presented a class-E controller to compensate for changes in coupling, comparing the transmit resonator voltage to a reference voltage and controlling the amplifier switching, with the goal of maintaining link efficiency with robustness to changes in coil separation. They also present a feedback control analysis of a coupled system for wireless electromagnetic power, and an experimental investigation of the effects of coil separation on peak efficiency. Experimental tests of their switching control system showed less than 16% variation in rectified output voltage over coil separations of 1–10 mm.

Ahn and Hong [34] adjusted operating frequency to maintain constant output voltage with coupling and load variations. The goal was to implement a low-power solution without requiring complex active circuits or external components. The switching frequency of a self-oscillating class-D power amplifier was controlled based on feedback from the drain of the switch transistor. The frequency-tuning system was demonstrated to have relatively constant output voltage over load variations and distances up to 12 mm, and able to maintain constant voltage over greater distances at higher load resistance.

Wang *et al.* [43] implemented a zero voltage switching follower design to compensate for coupling variations in inductively coupled systems, with the goal of providing power to an implantable heart pump. Switching frequency control was performed based on voltage feedback at the transmitter, in order to maintain middle zero voltage switching operation. The output power and efficiency were measured with and without frequency control, and the tuned system was able to deliver 10 W at greater coil separation and higher efficiency than without frequency control.

4.4. Impedance Tuning

Frequency tuning has been paired with automatic impedance matching to achieve resonance of an antenna while maintaining matching to a feed line. Van Schuylenbergh and Puers [40] adjusted the transmit frequency using a voltage controlled oscillator based on feedback of the dc input voltage of the internal regulator, communicated from the receiver using a third external sensing coil. Variable impedance matching at the transmit coil was also implemented based on the tuned transmit frequency, using a phase comparator to detect detuning of the coil. Hirata *et al.* [82] used two varactor diodes to tune the resonance frequency of a coil and to match the coil impedance to a transmission line. The matching was evaluated by sensing the phase change of reflected voltage at the transmitter.

Automatic impedance matching has also been implemented in systems operating at a fixed frequency to compensate for impedance variations that detune the system and decrease power transfer and efficiency. Impedance mismatches can be the result of antenna position or impedance changes due to tissue parameter variations or movement. It has been suggested that although frequency tuning is easier to accomplish, higher efficiencies are possible with impedance matching [68]. Adaptive matching and frequency tuning are two methods of adjusting the system to operate at a desired matching state; one adjusts the impedance of the system to achieve matching at a fixed operating frequency, while the other adjusts the frequency to operate at a matched system impedance. This is illustrated in Figure 5, with matching indicated as minimizing the reflection coefficient (Γ_{in}).



Figure 5. Illustrated relationship between impedance (Z) matching and frequency (f) tuning.

Si *et al.* [13,83] implemented switched capacitors to control effective capacitance in a push-pull resonant converter to achieve resonance at a given reference frequency, to compensate for changes due to load or circuit parameters. Rodes *et al.* [94] developed a tuning system for adjusting capacitance of a half-bridge voltage-mode resonant converter. The work referenced [13,83], stating that the half-bridge voltage mode resonant converter is better suited to medical applications due to the current-fed push-pull converter's inductances and the error terms in the switched capacitor transfer functions.

Waters *et al.* [84] demonstrated automatic impedance matching to match transmit and receive coil resonator input impedances to the source and load impedances, to accommodate changes in coupling of a magnetic resonant system due to changes in distance between the resonators. The demonstrated systems use extracted S-parameters to calculate the states of tunable matching networks. In the first case, the ideal matching impedance is calculated from the S-parameter matrix; in the second case, the matching state is found using an optimization algorithm, calculating the system efficiency from the S-parameters. The use of a network analyzer to extract S-parameters precludes direct application of this method in a real-time implantable system.

Chan Wai Po *et al.* developed a method of calculating complex antenna impedance based on detected voltage on capacitors, and using the calculated impedance to determine and set tunable reactive components in a matching network [31,80,89,93]. Implementations included MEMS variable inductors, switched capacitor networks, and varactors. The system was designed to compensate for impedance changes in pacemaker applications due to tissue differences, patient position, or nearby objects. Their one-iteration method reduces the time required for tuning, but requires vector calculations. Tuning power requirements are not provided, but it is proposed that the method reduces power requirements due to reduced tuning time. The method is applicable to both the transmitter and receiver, stated as the first system able to match both in a single process [31]. The transmit antenna impedance is matched to a power amplifier, while the receive antenna impedance is matched to a low noise amplifier.

Park and Ghovanloo [92] controlled for impedance variations due to the environment surrounding an intraoral sensing device. Switches in a CLC pi matching network were controlled by a microcontroller and set according to the output power monitored via a power detector. The system swept 16 possible settings and monitored output power, then set the configuration to the optimal switch setting, in a process taking 30 ms and repeated every second. However, the tuning was acknowledged to potentially impede real time operation due to interrupting communication of the intraoral device to an external receiver. The application was not specifically an implantable device, but was designed with size constraints to be worn on a dental retainer and addressed many of the same issues of energy transfer through tissue.

O'Driscoll *et al.* investigated the effects of misalignment, implantation depth, and tissue composition on system impedances in an implantable device, with focus on devices for neural recording [64,71,81,90]. The impedance was adjusted using a switched capacitor array, based on feedback of the voltage across the resistive load. The algorithm was a hybrid of gradient and binary search, where only the sign of the gradient was calculated to save power. The capacitor array and control were implemented on-chip, with the adaptation algorithm off-chip. The method is applicable to matching at both the transmitter and receiver.

Carta *et al.* [79] designed a self-tuning inductively coupled system to compensate for misalignments and distance changes, using a switched capacitor bank at the transmitter. The capacitor values were switched based on the voltage at the receiver, in a two stage process. First, the capacitance combinations were varied to determine the highest induced voltage at the receiver. Then, the capacitance was adjusted to maintain the voltage level determined in the first step.

Zargham and Gulak [72] developed a system for maximizing efficiency by tuning the load impedance to an optimal load. A switched capacitor array was controlled according to a sign-based gradient descent algorithm to maximize the power delivered to the rectifier load at the receiver. The implant receiver coil and circuitry were implemented on a single die in CMOS.

5. Summary and Future Directions

Miniature implantable devices are increasingly desirable to lessen the intrusiveness of the device for the patient and to reduce surgical complexity and infection risk. Electromagnetic energy transfer enables transcutaneous powering and communication with fully-implantable wireless medical devices, lessening the dependence on an implanted battery. The primary goal of transcutaneous energy transfer is to provide sufficient power to the implanted device while minimizing tissue heating due to absorbed energy. This has led to extensive research toward maximizing efficiency, through optimizing operating frequency, impedance matching and power delivery.

A significant complication of transcutaneous energy transfer is the variability of biological tissue in terms of tissue structure and electromagnetic properties. Tissue thickness and properties vary among individuals, among areas on the body and over time. Therefore, optimizing a transcutaneous system for a particular environment does not guarantee consistent operation in a real application.

Recent research has included adaptive functionality to compensate for variations encountered in the practical use of a transcutaneous system. These adaptive methods include tuning components of the system to maximize power to the implant within safety limits on absorption in tissue. Strategies at the external transmitter include power regulation, frequency tuning, beamforming and impedance matching to improve power transfer. Strategies at the receiver primarily involve impedance matching. Feedback provides closed loop control, including sensing at the receiver and communication to guide adjustments at the transmitter of power or frequency. A graphical summary of adaptive implementations in the literature is provided in Figure 6, including: input power (P_{in}) adjustment to compensate for changes in antenna distance (d) and alignment (m, θ_R); transmit antenna feed (field pattern E_T) adjustment to compensate for changes in implant antenna distance and alignment; frequency (f) and impedance (Z) tuning to compensate for changes in system impedances. Impedance tuning has been proposed at the external side (Tx) and/or the implant side (Rx) in several systems.



Figure 6. A graphical summary of tuning methods in the literature.

These adaptive methods are designed with regard to constraints on transcutaneous systems, primarily at the receiver, where tuning must be miniature and low power. Fewer restrictions exist at the transmitter in terms of size, but tuning time and complexity is still a concern. The goal of adaptation also depends on the characteristics of the system, mainly the field region as determined by the operating frequency, antenna sizes, antenna separation and properties of the transmission medium. The recent focus on miniature implants has led to research on focusing fields and midfield operation, achieving efficient power transfer in weakly coupled systems by minimizing tissue absorption, while accounting for implant migration and impedance variations. Challenges addressed in the literature include tradeoffs between tuning time and complexity, coordinating adaptation at the transmitter and receiver and implementing adaptation within power and size constraints at the implant.

Future directions for adaptive transcutaneous device research include addressing the effects of expected variations in the tissue medium, adaptation within power requirements of passive implants and monitoring absorption in tissue to inform adjustments to transmit power, frequency and impedance matching. Variations in the tissue medium are expected among patients, among different areas of the body and over time within a single patient and are therefore expected to contribute to variations in transcutaneous powering of implantable devices. Although progress has already been made on low-power adaptation of implanted devices [90], further efforts to reduce power requirements at the implant will enable passively-powered (battery-less) adaptive implants and, therefore, longer implant lifetime. To continue to address safety issues associated with transcutaneous power, future research will likely continue to address the adaptation of devices operating within SAR limitations. Specifically, methods of detecting SAR or tissue heating would provide another input for evaluating transcutaneous system efficiency in terms of power delivered relative to power dissipated in the tissue.

Implementations of adaptive transcutaneous systems will have substantial effects on implantable medical devices, enabling safer and reliable wireless powering and thereby reducing dependence on implanted batteries, as well as facilitating wireless and remote monitoring via implantable devices.

In particular, adaptive devices that are able to compensate for differences in the antenna positioning and system impedances have the potential to provide improved readings from implantable sensors due to more stable powering. Implantable medical devices have already revolutionized treatment and improved quality of life for many patients, and adaptive functionality is positioned to be the next major advance in implantable device technology.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

| ICD | Implantable cardioverter/defibrillator |
|------|---|
| VAD | Ventricular assist device |
| EM | Electromagnetic |
| FCC | Federal Communications Commission |
| SAR | Specific absorption rate |
| MEMS | Microelectromechanical systems |
| CMOS | Complementary metal-oxide semiconductor |
| Tx | Transmitter (external) |
| Rx | Receiver (implanted) |

References

- 1. Ratner, B.D.; Hoffman, A.S.; Schoen, F.J.; Lemons, J.E. *Biomaterials Science: An Introduction to Materials in Medicine*, 3rd ed.; Elsevier: Oxford, UK, 2013.
- 2. Medtronic. Protecta DR Implantable Defibrillator, 2013. Available online: www.medtronic.com (accessed on 4 January 2016).
- 3. St. Jude Medical. Current Plus ICD, 2013. Available online: www.sjm.com (accessed on 4 January 2016).
- 4. Medtronic. Adapta Pacemaker, 2015. Available online: www.medtronic.com (accessed on 4 January 2016).
- 5. Cyberonics. AspireSR, 2015. Available online: www.cyberonics.com (accessed on 4 January 2016).
- 6. Medtronic. SynchroMed II Pump, 2014. Available online: www.medtronic.com (accessed on 4 January 2016).
- 7. Cochlear Limited. Cochlear Implants, 2013. Available online: www.cochlear.com (accessed on 4 January 2016).
- 8. MED-EL. Cochlear Implants, 2013. Available online: www.medel.com (accessed on 4 January 2016).
- 9. Second Sight. The Argus II Retinal Prosthesis System, 2014. Available online: www.secondsight.com (accessed on 4 January 2016).
- Jonah, O.; Georgakopoulos, S.V.; Tentzeris, M.M. Orientation insensitive power transfer by magnetic resonance for mobile devices. In Proceedings of the IEEE Wireless Power Transfer Conference, Perugia, Italy, 15–16 May 2013; pp. 15–16.
- 11. Medtronic. Enlite Sensor, 2015. Available online: www.medtronicdiabetes.com (accessed on 4 January 2016).
- 12. Waters, B.H.; Smith, J.R.; Bonde, P. Innovative Free-Range Resonant Electrical Energy Delivery System (FREE-D System) for a Ventricular Assist Device Using Wireless Power. *ASAIO J.* **2014**, *60*, 31–37.
- 13. Si, P.; Hu, A.P.; Malpas, S.; Budgett, D. A Frequency Control Method for Regulating Wireless Power to Implantable Devices. *IEEE Trans. Biomed. Circuits Syst.* **2008**, *2*, 22–29.
- 14. Ng, D.C.; Bai, S.; Yang, J.; Tran, N.; Skafidas, E. Wireless technologies for closed-loop retinal prostheses. *J. Neural Eng.* **2009**, *6*, 065004.
- 15. Ho, J.S.; Yeh, A.J.; Neofytou, E.; Kim, S.; Tanabe, Y.; Patlolla, B.; Beygui, R.E.; Poon, A.S.Y. Wireless power transfer to deep-tissue microimplants. *Proc. Natl. Acad. Sci. USA* **2014**, *111*, 7974–7979.
- 16. Knecht, O.; Bosshard, R.; Kolar, J.W. High-Efficiency Transcutaneous Energy Transfer for Implantable Mechanical Heart Support Systems. *IEEE Trans. Power Electron.* **2015**, *30*, 6221–6236.
- 17. Amar, A.B.; Kouki, A.B.; Cao, H. Power Approaches for Implantable Medical Devices. *Sensors* 2015, 15, 28889–28914.

- Rabaey, J.M.; Mark, M.; Chen, D.; Sutardja, C.; Chongxuan, T.; Gowda, S.; Wagner, M.; Werthimer, D. Powering and communicating with mm-size implants. In Proceedings of the Design, Automation and Test in Europe Conference and Exhibition, Grenoble, France, 14–18 March 2011; pp. 1–6.
- Denisov, A.; Yeatman, E. Ultrasonic vs. Inductive Power Delivery for Miniature Biomedical Implants. In Proceedings of the International Conference on Body Sensor Networks (BSN), Singapore, 7–9 June 2010; pp. 84–89.
- 20. Roes, M.G.L.; Duarte, J.L.; Hendrix, M.A.M.; Lomonova, E.A. Acoustic Energy Transfer: A Review. *IEEE Trans. Ind. Electron.* 2013, 60, 242–248.
- 21. Brown, C.J.; Geers, A.; Herrmann, B.; Kirk, K.; Tomblin, J.B.; Waltzman, S. Cochlear Implants. Available online: www.asha.org/policy (accessed on 4 January 2016).
- 22. Advanced Bionics. The Harmony Sound Processor, 2015. Available online: www.advancedbionics.com (accessed on 4 January 2016).
- 23. Waters, B.H.; Sample, A.P.; Bonde, P.; Smith, J.R. Powering a Ventricular Assist Device (VAD) With the Free-Range Resonant Electrical Energy Delivery (FREE-D) System. *IEEE Proc.* **2012**, *100*, 138–149.
- 24. Ozeri, S.; Shmilovitz, D. Simultaneous backward data transmission and power harvesting in an ultrasonic transcutaneous energy transfer link employing acoustically dependent electric impedance modulation. *Ultrasonics* **2014**, *54*, 1929–1937.
- Mazzilli, F.; Kilinc, E.G.; Dehollain, C. 3.2 mW ultrasonic LSK modulator for uplink communication in deep implanted medical devices. In Proceedings of the IEEE Biomedical Circuits and Systems Conference (BioCAS), Lausanne, Switzerland, 22–24 October 2014; pp. 636–639.
- 26. Mazzilli, F.; Lafon, C.; Dehollain, C. A 10.5 cm ultrasound link for deep implanted medical devices. *IEEE Trans. Biomed. Circuits Syst.* **2014**, *8*, 738–750.
- 27. Poon, A.S.Y.; O'Driscoll, S.; Meng, T.H. Optimal frequency for wireless power transmission into dispersive tissue. *IEEE Trans. Antennas Propag.* 2010, *58*, 1739–1750.
- 28. Zargham, M.; Gulak, P.G. Fully Integrated on-Chip Coil in 0.13 μm CMOS for Wireless Power Transfer Through Biological Media. *IEEE Trans. Biomed. Circuits Syst.* **2014**, *9*, 259–271.
- 29. Hui, S.Y.R.; Zhong, W.; Lee, C.K. A Critical Review of Recent Progress in Mid-Range Wireless Power Transfer. *IEEE Trans. Power Electron.* **2014**, *29*, 4500–4511.
- 30. Zhang, Y.; Zhao, Z.; Chen, K. Frequency-Splitting Analysis of Four-Coil Resonant Wireless Power Transfer. *IEEE Trans. Ind. Appl.* **2014**, *50*, 2436–2445.
- Chan Wai Po, F.; De Foucauld, E.; David, J.B.; Delavaud, C.; Ciais, P. An Efficient Adaptive Antenna-Impedance Tuning Unit Designed for Wireless Pacemaker Telemetry. Available online: http://www.intechopen.com/books/modern-telemetry/an-efficient-adaptive-antenna-impedance-tuningunit-designed-for-wireless-pacemaker-telemetry (accessed on 4 January 2016).
- 32. Zargham, M.; Gulak, P.G. Maximum achievable efficiency in near-field coupled power-transfer systems. *IEEE Trans. Biomed. Circuits Syst.* **2012**, *6*, 228–245.
- 33. Li, H.; Li, J.; Wang, K.; Chen, W.; Yang, X. A Maximum Efficiency Point Tracking Control Scheme for Wireless Power Transfer Systems Using Magnetic Resonant Coupling. *IEEE Trans. Power Electron.* **2014**, *30*, 3998–4008.
- 34. Ahn, D.; Hong, S. Wireless Power Transmission With Self-Regulated Output Voltage for Biomedical Implant. *IEEE Trans. Ind. Electron.* **2014**, *61*, 2225–2235.
- Park, H.; Kiani, M.; Lee, H.M.; Kim, J.; Block, J.; Gosselin, B.; Ghovanloo, M. A Wireless Magnetoresistive Sensing System for an Intraoral Tongue-Computer Interface. *IEEE Trans. Biomed. Circuits Syst.* 2012, 6, 571–585.
- 36. Wang, G.; Liu, W.; Sivaprakasam, M.; Kendir, G.A. Design and analysis of an adaptive transcutaneous power telemetry for biomedical implants. *IEEE Trans. Circuits Syst. I Regul. Pap.* **2005**, *52*, 2109–2117.
- 37. Kiani, M.; Ghovanloo, M. An RFID-Based Closed-Loop Wireless Power Transmission System for Biomedical Applications. *IEEE Trans. Circuits Syst. II Express Br.* **2010**, *57*, 260–264.
- Christ, A.; Douglas, M.G.; Roman, J.M.; Cooper, E.B.; Sample, A.P.; Waters, B.H.; Smith, J.R.; Kuster, N. Evaluation of Wireless Resonant Power Transfer Systems With Human Electromagnetic Exposure Limits. *IEEE Trans. Electromagn. Compat.* 2013, 55, 265–274.
- Si, P.; Hu, A.P.; Hsu, J.W.; Chiang, M.; Wang, Y.; Malpas, S.; Budgett, D. Wireless Power Supply for Implantable Biomedical Device Based on Primary Input Voltage Regulation. In Proceeding of the 2nd IEEE Conference on Industrial Electronics and Applications, Harbin, China, 23–25 May 2007; pp. 235–239.

- 40. Van Schuylenbergh, K.; Puers, R. Self-tuning inductive powering for implantable telemetric monitoring systems. *Sens. Actuators A Phys.* **1996**, *52*, 1–7.
- Ng, D.C.; Felic, G.; Skafidas, E.; Bai, S. Closed-loop inductive link for wireless powering of a high density electrode array retinal prosthesis. In Proceedings of the Electromagnetic Compatibility Symposium Adelaide, Adelaide, Australia, 16–18 September 2009; pp. 92–97.
- 42. Kim, S.; Ho, J.S.; Poon, A.S.Y. Midfield Wireless Powering of Subwavelength Autonomous Devices. *Phys. Rev. Lett.* **2013**, *110*, 203905.
- 43. Wang, B.; Hu, A.P.; Budgett, D. Maintaining middle zero voltage switching operation of parallel-parallel tuned wireless power transfer system under bifurcation. *IET Power Electron.* **2014**, *7*, 78–84.
- 44. Schepps, J.L.; Foster, K.R. The UHF and microwave dielectric properties of normal and tumour tissues: Variation in dielectric properties with tissue water content. *Phys. Med. Biol.* **1980**, *25*, 1149–1159.
- Schwan, H.P. Electrical properties of tissues and cell suspensions: Mechanisms and models. In Proceedings of the 16th Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Baltimore, MD, USA, 3–6 November 1994; pp. A70–A71.
- 46. Gabriel, C.; Gabriel, S.; Corthout, E. The dielectric properties of biological tissues: I. Literature survey. *Phys. Med. Biol.* **1996**, *41*, 2231–2249.
- 47. Gabriel, S.; Lau, R.W.; Gabriel, C. The dielectric properties of biological tissues: II. Measurements in the frequency range 10 Hz to 20 GHz. *Phys. Med. Biol.* **1996**, *41*, 2251–2269.
- 48. Peyman, A.; Rezazadeh, A.A.; Gabriel, C. Changes in the dielectric properties of rat tissue as a function of age at microwave frequencies. *Phys. Med. Biol.* **2001**, *46*, 1617–1629.
- Christ, A.; Klingenbock, A.; Samaras, T.; Goiceanu, C.; Kuster, N. The dependence of electromagnetic far-field absorption on body tissue composition in the frequency range from 300 MHz to 6 GHz. *IEEE Trans. Microw. Theory Tech.* 2006, 54, 2188–2195.
- 50. Kanehisa, H.; Miyatani, M.; Azuma, K.; Kuno, S.; Fukunaga, T. Influences of age and sex on abdominal muscle and subcutaneous fat thickness. *Eur. J. Appl. Physiol.* **2004**, *91*, 534–537.
- 51. Peyman, A.; Gabriel, C.; Grant, E.H.; Vermeeren, G.; Martens, L. Variation of the dielectric properties of tissues with age: The effect on the values of SAR in children when exposed to walkie-talkie devices. *Phys. Med. Biol.* **2009**, *54*, 227–241.
- Hu, S.; Hao Cheong, J.; Gao, Y.; Ho, M.C.K.; Je, M.; Madihian, M.; Khait, S. A low-cost 2.45-GHz wireless power link for biomedical devices. In Proceedings of the IEEE Asia-Pacific Conference on Antennas and Propagation, Singapore, Singapore, 27–29 August 2012; pp. 215–216.
- Cleveland, R.F., Jr.; Sylvar, D.M.; Ulcek, J.L. Evaluating Compliance with FCC Guidelines for Human Exposure to Radiofrequency Electromagnetic Fields. Available online: https://transition.fcc.gov/ Bureaus/Engineering_Technology/Documents/bulletins/oet65/oet65b.pdf (accessed on 4 January 2016).
- 54. Balanis, C.A. Antenna Theory: Analysis and Design; John Wiley and Sons: Hoboken, NJ, USA, 2005.
- 55. Liu, X.; Berger, J.L.; Ogirala, A.; Mickle, M.H. A touch probe method of operating an implantable RFID tag for orthopedic implant identification. *IEEE Trans. Biomed. Circuits Syst.* **2013**, *7*, 236–242.
- 56. Waters, B.H.; Reed, J.T.; Kagi, K.R.; Sample, A.P.; Bonde, P.; Smith, J.R. A portable transmitter for wirelessly powering a ventricular assist device using the free-range resonant electrical energy delivery (FREE-D) system. In *Wirelessly Powered Sensor Networks and Computational RFID*; Smith, J.R., Ed.; Springer: New York, NY, USA, 2013; pp. 235–247.
- 57. Ho, J.S.; Sanghoek, K.; Poon, A.S.Y. Midfield wireless powering for implantable systems. *IEEE Proc.* **2013**, 101, 1369–1378.
- 58. Gabriel, S.; Lau, R.W.; Gabriel, C. The dielectric properties of biological tissues: III. Parametric models for the dielectric spectrum of tissues. *Phys. Med. Biol.* **1996**, *41*, 2271–2293.
- 59. *IEEE Standard for Safety Levels with Respect to Human Exposure to Radio Frequency Electromagnetic Fields,* 3 kHz to 300 GHz; Institute of Electrical and Electronics Engineers: New York, NY, USA, 2005.
- 60. International Commission on Non-Ionizing Radiation Protection. Guidelines for Limiting Exposure to Time-Varying Electric, Magnetic, and Electromagnetic Fields (up to 300 GHz). *Health Phys.* **1998**, *74*, 494-522.
- 61. Cleveland, R.F., Jr.; Ulcek, J.L. Questions and Answers about Biological Effects and Potential Hazards of Radiofrequency Electromagnetic Fields; Federal Communications Commission: Washington, DC, USA, 1999.
- 62. Kim, S.; Tathireddy, P.; Normann, R.A.; Solzbacher, F. Thermal impact of an active 3-D microelectrode array implanted in the brain. *IEEE Trans. Neural Syst. Rehabil. Eng.* **2007**, *15*, 493–501.

- 63. Kyriakou, A.; Christ, A.; Neufeld, E.; Kuster, N. Local tissue temperature increase of a generic implant compared to the basic restrictions defined in safety guidelines. *Bioelectromagnetics* **2012**, *33*, 366–374.
- O'Driscoll, S.; Poon, A.S.Y.; Meng, T.H. A mm-sized implantable power receiver with adaptive link compensation. In Proceedings of the IEEE International Solid-State Circuits Conference, San Francisco, CA, USA, 8–12 February 2009; pp. 294–295.
- 65. Ziolkowski, R.W.; Erentok, A. Metamaterial-based efficient electrically small antennas. *IEEE Trans. Antennas Propag.* **2006**, *54*, 2113–2130.
- Poon, A.S.Y. A General Solution to Wireless Power Transfer Between Two Circular Loop. *Prog. Electromagn. Res.* 2014, 148, 171–182.
- Mark, M.; Bjorninen, T.; Ukkonen, L.; Sydanheimo, L.; Rabaey, J.M. SAR reduction and link optimization for mm-size remotely powered wireless implants using segmented loop antennas. In Proceedings of the IEEE Topical Conference on Biomedical Wireless Technologies, Networks, and Sensing Systems, Phoenix, AZ, USA, 16–19 January 2011; pp. 7–10.
- Heebl, J.D.; Thomas, E.M.; Penno, R.P.; Grbic, A. Comprehensive analysis and measurement of frequency-tuned and impedance-tuned wireless non-radiative power transfer systems. *IEEE Antennas Propag. Mag.* 2014, 56, 44–60.
- 69. Lee, T.H. *The Design of CMOS Radio-Frequency Integrated Circuits;* Cabridge University Press: New York, NY, USA, 2004.
- 70. Luo, Y.; Yang, Y.; Chen, Z. Self-tuning Wireless Power Transmission Scheme Based on on-line Scattering Parameters Measurement and Two-side Power Matching. *Sci. Rep.* **2014**, *4*, 4332.
- 71. O'Driscoll, S. Adaptive Signal Acquisition and Power Delivery for Implanted Medical Devices. Ph.D. Thesis, Stanford University, Stanford, CA, USA, 2009.
- Zargham, M.; Gulak, P.G. A 0.13 μm CMOS integrated wireless power receiver for biomedical applications. In Proceedings of the ESSCIRC, Bucharest, Romania, 16–20 September 2013; pp. 137–140.
- 73. Baker, M.W.; Sarpeshkar, R. Feedback Analysis and Design of RF Power Links for Low-Power Bionic Systems. *IEEE Trans. Biomed. Circuits Syst.* **2007**, *1*, 28–38.
- 74. Pozar, D.M. Microwave Engineering, 3rd ed.; John Wiley and Sons: Hoboken, NJ, USA, 2005.
- 75. Yakovlev, A.; Sanghoek, K.; Poon, A. Implantable Biomedical Devices: Wireless powering and communication. *IEEE Commun. Mag.* 2012, *50*, 152–159.
- Ahn, D.; Hong, S. Wireless Power Transfer Resonance Coupling Amplification by Load-Modulation Switching Controller. *IEEE Trans. Ind. Electron.* 2015, 62, 898–909.
- 77. Ko, W.H.; Liang, S.P. RF-Powered Cage System for Implant Biotelemetry. *IEEE Trans. Biomed. Eng.* **1980**, *BME-27*, 460–467.
- 78. Fernald, K.W.; Paulos, J.J.; Stackhouse, B.A.; Heaton, R.A. A self-tuning digital telemetry IC for use in a microprocessor-based implantable instrument. *IEEE J. Solid-State Circuits* **1992**, 27, 1826–1832.
- 79. Carta, R.; Thoné, J.; Gosset, G.; Cogels, G.; Flandre, D.; Puers, R. A Self-Tuning Inductive Powering System for Biomedical Implants. *Procedia Eng.* **2011**, *25*, 1585–1588.
- Chan Wai Po, F.; De Foucauld, E.; Delavaud, C.; Ciais, P.; Kerherve, E. An vector automatic matching system designed for wireless medical telemetry. In Proceedings of the Joint 6th International IEEE Northeast Workshop on Circuits and Systems and TAISA Conference, Montreal, QC, Canada, 22–25 June 2008; pp. 89–92.
- 81. O'Driscol, S.D. A mm-sized implantable power receiver with adaptive matching. In Proceedings of the IEEE Sensors, Waikoloa, HI, USA, 1–4 November 2010; pp. 83–88.
- 82. Hirata, H.; Walczak, T.; Swartz, H.M. Electronically Tunable Surface-Coil-Type Resonator for L-Band EPR Spectroscopy. J. Magn. Reson. 2000, 142, 159–167.
- Si, P.; Hu, A.P.; Budgett, D.; Malpas, S.; Yang, J.; Gao, J. Stabilizing the operating frequency of a resonant converter for wireless power transfer to implantable biomedical sensors. In Proceedings of the 1st International Conference on Sensing Technology, Palmerston North, New Zealand, 21–23 November 2005; pp. 477–482.
- Waters, B.H.; Sample, A.P.; Smith, J.R. Adaptive impedance matching for magnetically coupled resonators. In Proceedings of the Progress in Electromagnetics Research Symposium, Kuala Lumpur, Malaysia, 27–30 March 2012.
- 85. Sjoblom, P.; Sjoland, H. Measured CMOS Switched High-Quality Capacitors in a Reconfigurable Matching Network. *IEEE Trans. Circuits Syst. II Express Br.* **2007**, *54*, 858–862.

- 86. Kiani, M.; Jow, U.M.; Ghovanloo, M. Design and Optimization of a 3-Coil Inductive Link for Efficient Wireless Power Transmission. *IEEE Trans. Biomed. Circuits Syst.* **2011**, *5*, 579–591.
- 87. Cetiner, B.A.; Qian, J.Y.; Li, G.P.; De Flaviis, F. A reconfigurable spiral antenna for adaptive MIMO systems. *EURASIP J. Wirel. Commun. Netw.* **2005**, 2005, 382–389.
- 88. Shmilovitz, D.; Ozeri, S.; Wang, C.C.; Spivak, B. Noninvasive Control of the Power Transferred to an Implanted Device by an Ultrasonic Transcutaneous Energy Transfer Link. *IEEE Trans. Biomed. Eng.* **2014**, *61*, 995–1004.
- Chan Wai Po, F.; De Foucauld, E.; Vincent, P.; Hameau, F.; Kerherve, E.; Dal Molin, R.; Pons, P.; Pierquin, R. A 2.4 GHz CMOS automatic matching network design for pacemaker applications. In Proceedings of the Joint IEEE North-East Workshop on Circuits and Systems and TAISA Conference, Toulouse, France, 28 June–1 July 2009; pp. 1–4.
- 90. O'Driscoll, S.; Meng, T.H. Adaptive signal acquisition and wireless power transfer for an implantable prosthesis processor. In Proceedings of the IEEE International Symposium on Circuits and Systems, Paris, France, 30 May–2 June 2010; pp. 3589–3592.
- 91. Sample, A.P.; Waters, B.H.; Wisdom, S.T.; Smith, J.R. Enabling Seamless Wireless Power Delivery in Dynamic Environments. *IEEE Proc.* **2013**, *101*, 1343–1358.
- 92. Park, H.; Ghovanloo, M. Wireless Communication of Intraoral Devices and Its Optimal Frequency Selection. *IEEE Trans. Microw. Theory Tech.* **2014**, *62*, 3205–3215.
- 93. Chan Wai Po, F.; De Foucauld, E.; Vincent, P.; Hameau, F.; Morche, D.; Delaveaud, C.; Dal Molin, R.; Pons, P.; Pierquin, R.; Kerherve, E. A fast and accurate automatic matching network designed for ultra low power medical applications. In Proceedings of the IEEE International Symposium on Circuits and Systems, Taipei, Taiwan, 24–27 May 2009; pp. 673–676.
- 94. Rodes, F.; Zhang, M.; Denieport, R.; Wang, X. Optimization of the Power Transfer Through Human Body with an Auto-Tuning System Using a Synchronous Switched Capacitor. *IEEE Trans. Circuits Syst. II Express Br.* **2015**, *62*, 129–133.
- 95. Danilov, A.A.; Mindubaev, E.A. Influence of Angular Coil Displacements on Effectiveness of Wireless Transcutaneous Inductive Energy Transmission. *Biomed. Eng.* **2015**, *49*, 171–173.
- 96. McMenamin, P.; Uei-Ming, J.; Kiani, M.; Manns, J.R.; Ghovanloo, M. A smart cage for behavioral experiments on small freely behaving animal subjects. In Proceedings of the 6th International IEEE/EMBS Conference on Neural Engineering, San Diego, CA, USA, 6–8 November 2013; pp. 985–988.
- 97. Hannan, M.A.; Hussein, H.A.; Mutashar, S.; Samad, S.A.; Hussain, A. Automatic Frequency Controller for Power Amplifiers Used in Bio-Implanted Applications: Issues and Challenges. *Sensors* **2014**, *14*, 23843–23870.



© 2016 by the authors; licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons by Attribution (CC-BY) license (http://creativecommons.org/licenses/by/4.0/).